Author's response to reviews

Title: Effectiveness of oral nutritional supplementation for older women after a fracture: a randomized controlled pilot study

Authors:

Ian D Cameron (ianc@mail.usyd.edu.au)
Susan E Kurrle (kurrle@bigpond.com)
Cesar Uy (cuy@nsccahs.health.nsw.gov.au)
Keri A Lockwood (kalockwood@nsccahs.health.nsw.gov.au)
Lydia Au (Lydia_AU@Alexhosp.com.sg)
Frederieke G Schaafsma (f.g.schaafsma@gmail.com)

Version: 5 Date: 8 June 2011

Author's response to reviews: see over
Natasha Mellins-Cohen  
BMC-series Journals  
BioMed Central  
Floor 6, 236 Gray's Inn Road,  
London, WC1X 8HL  

Regarding: Review 953018152376952  

Ryde, 08-06-2011  

Dear Natasha Mellins-Cohen,  

Thank you for accepting our manuscript for publication. We have addressed the necessary changes to the format of our manuscript, so that the published article conforms to your journal style. We hope that with these last modifications the paper will be ready for publication.  

Kind regards,  

Professor Ian Cameron  
Rehabilitation Studies Unit  
Sydney Medical School  
University of Sydney  
PO Box 6, Ryde NSW 1680
Formatting changes to be made:

1. Title: please ensure that the title is identical in the manuscript and the submission system.
   
   Done.

2. Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is 'First name Initial Last name' (e.g. Joe F Bloggs).
   
   Done

3. Abstract: This should be identical in the manuscript and the submission system.
   
   Done

4. Structure: Please make 'Conclusion' a heading.
   
   Done

5. References: reference 17 contains the details for multiple sources, please separate the details into separate reference details, and update the text accordingly, so that each source is cited correctly.
   
   Done

6. References: Please change 'Appenix A' to 'Appendix 1'.
   
   Done

7. Tables- please remove the visible vertical lines from your tables.
   
   Done

8. Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.
   
   Done